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 BEAM: 
 Shutdown started Fri 9/5 more than 6 weeks,  
 Expecting  NuMI 10/27 

 NearDet:  
 It is up and running as cool, with full HV. 
 Blackened out by Velcro.  
 Taking special runs. 
 

 FarDet:  
 It is up and running with 2 partitions.  
 Physics (cosmics) 12-diblocks running. 
 Retrofitting is spread over through only 2 diblocks. 
 Taking special runs. 
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Operations I – summary  



FarDet: Data-Taking 
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Partition 1 (status last week) 
 diblocks 3 – 14  
 Running ~99% active,  
 ALL COOLED, HV at full gain. 
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Current status - Retrofitting: 
 Retrofit: ALL OLD APDs & brass fittings away! 
 Including 20 APDs on DiB-06. 
 Already retrofitted DiB-{04,05,03}. 
 New APDs installed on DiB-02, and then 01!  
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FarDet – recent status: 
 Physics partition contains 12 DiBs in one stack (03-14).  
 Special runs started: 
 Lights ON/OFF at FarDet 
 Multi-point readout with 8 points/hit to see time resolution and compare 

with 4 points/hit. 
 “Multi-gain” – setup different gains, stepped up 100 to 156 for single 

diblocks, check capability of DAQ system and noisiness of detector. 
 Trigger stress runs & Supernova trigger accommodation being tested.  

 Installing new APDs on retrofitted parts, now waiting on producers. 
 Also several checks and tightening of the Dry Gas systems done. 

Shown as very helpful to stabilize detector (several APDs was found 
and swapped). 

Operations II - FarDet 



 NearDet:  
 Lights ON/OFF issue, part of detector very sense to  
 blackened out, helped but still sensitive. 
 Tests for multipoint readout for nanoslice version 2: 
    8 instead of 4 points in readout - done. 
 In preparation tests for various raising/falling times (FEB electronics) to 

confirm if we can improve the time resolution. 
 ‘Big’ maintenance of the detector: 29 APDs were swapped. The main 

goal is to lower the noisiness. Except 4 parts all satisfied. 
 There is today the attempt to check oil filling status. Based on the 

measurement results a strategy for next oil top-up will be figured out. 
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Operations III - NearDet 
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ROC-West: 
 MOVED IN 2 weeks ago, all shifts are being done here! 
 Last weekend full operation during the WH power outage! 
 Main DAQ and DCS computers ready. 
 Phone connection established 
 Some issues pending: 
Card reader 
Underground key box 
 iFix computer! 
 Video connection 

to Remote Control 
rooms - tested 

Operations IV – ROC-West  
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Operations V – ROC-West  shift 
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